Author’s response to reviews

Title: Implementing the EffTox Dose-Finding Design in the Matchpoint Trial

Authors:

Kristian Brock (k.brock@bham.ac.uk;kristian.brock@gmail.com)
Lucinda Billingham (L.J.Billingham@bham.ac.uk)
Mhairi Copland (Mhairi.Copland@glasgow.ac.uk)
Shamyla Siddique (S.Siddique@bham.ac.uk)
Mirjana Sirovica (M.Sirovica@bham.ac.uk)
Christina Yap (C.Yap@bham.ac.uk)

Version: 2 Date: 11 Jun 2017

Author’s response to reviews:

Authors' responses to editor's comments on BMC Medical Research manuscript BMRM-D-17-00049 "Implementing the EffTox Dose-Finding Design in the Matchpoint Trial" by Brock et al.

Dear Sir / Madam,

Thank you for pointing out the inappropriate reference in the final paragraph of the "EffTox in the Matchpoint trial" section. We have rectified this and attach a new draft manuscript.

Yours sincerely,

Kristian Brock

Cancer Research UK Clinical Trials Unit, University of Birmingham, UK